Author's response to reviews

Title: Theories of schizophrenia: a genetic-inflammatory-vascular synthesis

Authors:

Daniel R. Hanson (drhanson@umn.edu)
Irving I. Gottesman (gotte003@umn.edu)

Version: 3 Date: 9 February 2005

Author's response to reviews:

There were no changes suggested by the reviewer

All of the recommended formatting changes were made

The abstract was shortened to 347 words and restructured according to directions

The competing interests statement was changed to conform to the recommendations

The figure captions were renamed using numerals

The reference section provides the foundation and validation for all of our assertions and for our hypothesis. It would damage the scholarship of the paper to reduce these and we beg you indulgence in this respect.

Thanks

Daniel R Hanson, MF, PhD